Author’s response to reviews

Title: Internal Structure and Reliability of the Attachment Insecurity Screening Inventory (AISI) for Children Age 6 to 12

Authors:

Anouk Spruit (a.spruit@uva.nl)

Inge Wissink (i.b.wissink@uva.nl)

Marc Noom (m.j.noom@uva.nl)

Cristina Colonnesi (c.colonnesi@uva.nl)

Nelleke Polderman (nelleke.polderman@basictrust.com)

Lucia Willems (l.d.willems@uva.nl)

Charlotte Barning (cbarning@wsg.nu)

Geert Jan Stams (g.j.j.m.stams@uva.nl)

Version: 2 Date: 04 Jan 2018

Author’s response to reviews:

All instructions have been followed.